Author's response to reviews

Title: Decreased PCSK9 expression in human Hepatocellular Carcinoma

Authors:

Mamatha Bhat (mamatha.bhat@mcgill.ca)
Nicolas Skill (nskill@iupui.edu)
Victoria Marcus (victoria.marcus@muhc.mcgill.ca)
Marc Deschenes (marc.deschenes@muhc.mcgill.ca)
Xianming Tan (xianming.tan@clinepi.mcgill.ca)
Jeanne Bouteaud (jeanne.bouteaud@mail.mcgill.ca)
Sarita Negi (sarita.negi@gmail.com)
Zuhier Awan (zuhier_awan@yahoo.com)
Reid Aikin (reid.aikin@mail.mcgill.ca)
Janet Kwan (janet.kwan@mail.mcgill.ca)
Ramila Amre (ramila.amre@muhc.mcgill.ca)
Sebastien Tabaries (sebastien.tabaries@mcgill.ca)
Mazen Hassanain (mazen.hassanain@mcgill.ca)
Nabil G. Seidah (nabil.seidah@ircm.qc.ca)
Mary Maluccio (mmaluccio@iupui.edu)
Peter Siegel (peter.siegel@mcgill.ca)
Peter Metrakos (peter.metrakos@mcgill.ca)

Version: 5
Date: 27 August 2015

Author's response to reviews:

Dear Editor,

Thank you for your e-mail confirming acceptance of our manuscript. Please find below our point-by-point response to your requested revisions.

1) Please provide the full postal address for the corresponding author.
This has been added to the title page.

2) Please provide a heading for the Conclusions section.
This heading has been added.

3) Please upload all figures as separate figure files and remove them from the main manuscript. Figure legends should, however, be placed at the end of the main manuscript.
I have uploaded the figures separately, and removed them from the main manuscript.

4) Please place figure legends and tables after the References.
The figure legends and tables have been placed after the references.
Thank you for your interest in our manuscript.
Best regards,
Mamatha Bhat, MD FRCPC
McGill University Health Centre